\documentclass[english,a4paper]{article}
\input{crypto_preamble.tex}
\usepackage{hyperref} % needed for chunk1
\usepackage{boxedminipage} % needed for chunk3
\begin{document}

\handout{0368.4162: Introduction to Cryptography}
        {Ran Canetti}
        {03 November 2008}
        {Fall 2008}
        {Lecture 1} % title
        {D{}. Shahaf, D{}. Sotnikov, N{}. Bitansky} % in order of our parts

\input{crypto01_chunk1.tex}
\input{crypto01_chunk2.tex}
\input{crypto01_chunk3.tex}

\end{document}

